Author's response to reviews

Title: Identification and Regulation of c-Myb Target Genes in MCF-7 Cells

Authors:

Anita M. Quintana (amquintana@salud.unm.edu)
Fan Liu (liuf1@mskcc.org)
John P. O'Rourke (JORourkeJr@salud.unm.edu)
Scott A. Ness (ness@unm.edu)

Version: 2 Date: 16 August 2010

Author's response to reviews:

This manuscript was submitted to Breast Cancer Research, but the editors thought it better suited for BMC Cancer.

The manuscript details the use of global chromatin immunoprecipitation assays to identify target genes for the c-Myb oncoprotein in human breast cancer cells (MCF-7 cell line). This is important information that will benefit many researchers.